Author's response to reviews

Title: Levels and determinants of completing the continuum of care for maternal and newborn health in Cambodia—evidence from a population-based survey

Authors:

Wenjuan Wang (wenjuan.wang@icfi.com)
Rathavuth Hong (rhong@icfi.com)

Version: 2 Date: 7 January 2014

Author's response to reviews: see over
January 7, 2014

Janelyn Ann Cruz
BMC Pregnancy and Childbirth

Dear Ms. Cruz,

Thank you very much for the helpful comments concerning our paper, "Levels and determinants of completing the continuum of care for maternal and newborn health in Cambodia-evidence from a population-based survey"

We have been able to revise this paper and are submitting it again for your review. Each of your suggestions and comments has been addressed, and our response follows.

Comment #1:

Competing Interests
Please be advised that manuscripts must include a ‘Competing interests’ section. This should be placed after the Conclusions/Abbreviations. If there are none to declare, please include the statement “The authors declare that they have no competing interests.” Please consider the following questions and include an appropriate declaration of competing interests in your manuscript:

A statement on competing interests has been added in after the Conclusions in the resubmitted main manuscript

Comments #2:

Ethics
Research involving human subjects, human material, or human data, must have been performed in accordance with the Declaration of Helsinki (http://www.wma.net/en/30publications/10policies/b3/index.html) and must have been approved by an appropriate ethics committee. Experimental research on vertebrates or any regulated invertebrates must comply with institutional, national, or international guidelines, and where available should have been approved by an appropriate ethics committee. A statement detailing what ethical approval has been obtained, including the name of the ethics committee and the reference number where appropriate, must appear in all manuscripts. If a study has been granted an exemption from requiring ethics approval or does not require approval, this should also be detailed in the manuscript (including the name of the ethics committee that granted the exemption). For all research involving human subjects, informed consent to participate in the study should be obtained from participants (or their parent or guardian in the case of children under 16) and a statement to this effect should appear in the manuscript. Please refer to our Editorial Policies for full details of the ethical requirements for the journal: http://www.biomedcentral.com/about/editorialpolicies#Ethics.

A statement that details the ethical approval has been added before the newly added Competing Interests in the resubmitted manuscript.